Author’s response to reviews

Title: Acanthamoeba keratitis related to contact lens use in a tertiary hospital in China

Authors:
Weiwei Li (lww994012@163.com)
Zhiqun Wang (eyewzq@163.com)
Jinghao Qu (qujinghaochn@hotmail.com)
Yang Zhang (biozy1@126.com)
Xuguang Sun (sunxg1955@163.com)

Version: 3 Date: 14 Aug 2019

Author’s response to reviews:

Dear editor,

We greatly appreciate the help from you and all of the reviewers concerning about reviewing of the manuscript “Acanthamoeba keratitis related to contact lens in a tertiary hospital in China”. We revised the manuscript according to the editors’ comments. The answers to the questions from the editors and reviewers are as follows:

Comments from the editor:

Comment 1: Please ensure the corresponding author has been indicated on the title page.
Answer: The corresponding author has been marked on the title page.

Comment 2: Please insert the heading 'Declarations' for the Declarations section, above the "Authors' contributions" section of your manuscript.
Answer: Declarations section has been inserted before the authors’ contributions section.

Comment 3: In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.
Answer: Written informed consents were obtained from all patients participating in the investigation.

Comment 4: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

Answer: Manuscript have been revised according to the editor’s comments.

The authors thank the editor for the invaluable expertise and assistance.

Sincerely yours,

Xuguang Sun